Author's response to reviews

Title: Comparison of two recombinant erythropoietin formulations in patients with anemia due to end-stage renal disease on hemodialysis: A parallel, randomized, double blind study.

Authors:

Jorge F Perez-Oliva (jpolivd@infomed.sld.cu)
Martha Casanova-Gonzalez (martahaf@jagua.cfg.sld.cu)
Idrian Garcia-Garcia (idrian.garcia@cigb.edu.cu)
Pedro J Porrero-Martín (pedro.porrero@cigb.edu.cu)
Carmen M Valenzuela-Silva (carmen.valenzuela@cigb.edu.cu)
Tairí Hernandez-Montero (tairi.hernandez@cnic.edu.cu)
Marcia Lagarde-Ampudia (marciala@infomed.sld.cu)
Yuri Casanova-Kutsareva (yuri.casanova@cigb.edu.cu)
Yisel Avila-Albuene (yisel@cencec.sld.cu)
Alicia Vargas-Batista (alicia@cencec.sld.cu)
Hailen Bobillo-Lopez (hailen.bobillo@cigb.edu.cu)
Raul Herrera-Valdes (insnef@infomed.sld.cu)
Pedro A Lopez-Saura (lopez.saura@cigb.edu.cu)

Version: 4 Date: 21 May 2005

Author's response to reviews: see over
May 21, 2005

Jane M Borthwick PhD
Assistant Editor
BMC Journals
BioMed Central Ltd.

Manuscript: Comparison of two recombinant erythropoietin formulations in patients with anemia due to end-stage renal disease on hemodialysis: A parallel, randomized, double blind study

Dear Jane M Borthwick:

The manuscript was carefully checked. The formatting changes requested are answered below.

Author list - The list of authors in the manuscript is written identically and in the order in which they are listed in the submission system. None changes were necessary.

Abstract - The background section was improved. The corresponding sentences with an introduction to the subject and an overview of the aims of the research were added.

Affiliations - Each affiliation was written in full, provided in the required format: department, institute, city, country.

Each author was linked to their corresponding affiliation with the use of superscript numbers.

Section headings - Sentence case was applied for all headings and sub-headings in the manuscript. All unnecessary capitals were removed.

Figures - The figures were cropped as closely as possible to minimize white space around the image.

Sincerely,

Dr. Idrian Garcia-Garcia

Center for Biological Research
Clinical Trials Division,
PO Box 6332, Havana, CUBA
Telephone: +53 (7) 2087379; 2087465; Fax: +53 (7) 2736008
Email: idrian.garcia@cigb.edu.cu